Author’s response to reviews

Title: Development of a Management Algorithm for Post-operative Pain (MAPP) after total knee or total hip replacement: Study rationale and design.

Authors:

Mari Botti (mari.botti@deakin.edu.au)
Bridie Kent (bridie.kent@plymouth.ac.uk)
Tracey Bucknall (tracey.bucknall@deakin.edu.au)
Maxine Duke (maxine.duke@deakin.edu.au)
Megan-Jane Johnstone (megan-jane.johnstone@deakin.edu.au)
Julie Considine (julie.considine@deakin.edu.au)
Bernice Redley (bernice.redley@deakin.edu.au)
Susan Hunter (susan.hunter@deakin.edu.au)
Richard de Steiger (Richard.DeSteiger@epworth.org.au)
Marlene Holcombe (marlene.holcombe@epworth.org.au)
Emma Cohen (emma.cohen@deakin.edu.au)

Version: 4 Date: 2 July 2014

Author’s response to reviews:

To the Editor,

Please find the completed CONSORT RCT checklist attached to this submission.

Kind regards,

Emma